



Article Reconfigurable Intelligent Surface (RIS)-Assisted Non-Terrestrial Network (NTN)-Based 6G Communications: A Contemporary Survey

Chika E. Worka¹, Faheem A. Khan¹, Qasim Zeeshan Ahmed¹, Pradorn Sureephong^{2,*} and Temitope Alade³

- ¹ Department of Computing and Engineering, University of Huddersfield, Huddersfield HD1 3DH, UK; chika.worka@hud.ac.uk (C.E.W.); f.khan@hud.ac.uk (F.A.K.); q.ahmed@hud.ac.uk (Q.Z.A.)
- ² College of Arts, Media and Technology, Chiang Mai University, Chiang Mai 50200, Thailand
- ³ School of Computing Sciences, University of East Anglia, Norwich Research Park, Norwich NR4 7TJ, UK; t.alade@uea.ac.uk
- * Correspondence: pradorn.s@cmu.ac.th

Abstract: This article examines the transformative potential of integrating reconfigurable intelligent surfaces (RISs) into sixth-generation (6G) wireless non-terrestrial networks (NTNs). The focus is on the RIS's capability to address diverse user requirements, including secure data transmission, power efficiency, extended coverage, and enhanced data rates. The paper delves into the synergy between RISs and NTNs, emphasizing key components like multiple-input multiple-output (MIMO) systems and advanced radio communications. Additionally, it highlights the crucial role of artificial intelligence (AI) and machine learning (ML) in optimizing RIS-based beamforming to solve scientific and engineering challenges while ensuring energy efficiency and sustainability in NTN operations. By positioning RISs as a key enabler in shaping the future of wireless communication systems, this research underscores their significance in unlocking the full potential of NTNs and advancing next-generation wireless communications. This paper contributes valuable insights and projections for future research directions, highlighting RISs' potential to revolutionize NTNs for 6G technologies.



1. Introduction

High-altitude platforms (HAPs), such as satellites and unmanned aerial vehicles (UAVs), are critical components of non-terrestrial networks (NTNs). HAPs operate at altitudes exceeding 20 km, surpassing traditional terrestrial networks (TNs), and possess the unique ability to transcend geographical barriers. As essential elements of NTNs, HAPs drive the advancement of wireless communication beyond terrestrial constraints, promising comprehensive coverage, enhanced connectivity, and adaptable functionality [1–4]. They are crucial in bridging the communication gap between remote or challenging terrains and enable diverse applications, ranging from aerial base stations (BSs) to cellular-connected NTNs [5–9].

In wireless communications, HAPs offer a strategic advantage due to their superior signal propagation capabilities and expanded coverage. This unique spatial positioning, combined with the flexibility of UAVs, makes HAPs dynamic and adaptive assets in the wireless communication arsenal [10]. Their deployment enriches NTN architectures, meeting the growing demand for connectivity in rural, oceanic, and challenging environments where terrestrial stations struggle to perform optimally [11]. This transformative impact of HAPs helps to address key challenges faced by terrestrial BSs. In scenarios with inadequate or compromised terrestrial infrastructure, HAPs offer agile and efficient solutions, ensuring uninterrupted connectivity. Their mobility and adaptability are particularly beneficial in



Citation: Worka, C.E.; Khan, F.A.; Ahmed, Q.Z.; Sureephong, P.; Alade, T. Reconfigurable Intelligent Surface (RIS)-Assisted Non-Terrestrial Network (NTN)-Based 6G Communications: A Contemporary Survey. *Sensors* **2024**, *24*, 6958. https://doi.org/10.3390/s24216958

Academic Editors: Tiejun Lv, Jie Zeng and Xin Su

Received: 5 September 2024 Revised: 21 October 2024 Accepted: 28 October 2024 Published: 30 October 2024



Copyright: © 2024 by the authors. Licensee MDPI, Basel, Switzerland. This article is an open access article distributed under the terms and conditions of the Creative Commons Attribution (CC BY) license (https:// creativecommons.org/licenses/by/ 4.0/). emergencies, crowded events, or remote areas [12]. By serving as aerial base stations (BSs), HAPs redefine the concept of on-demand assistance to cellular networks, overcoming limitations associated with traditional TNs.

The dynamic and unpredictable nature of non-terrestrial BSs underscores the need for advanced coverage optimization solutions, highlighting the critical role of HAPs in the NTN landscape [13]. Machine learning (ML) algorithms are used in contrast to traditional mathematical optimization methods. This approach has become essential for efficiently managing resources and mobility in HAP stations. HAPs, acting as intelligent agents, leverage ML techniques within Markov Decision Processes to optimize trajectories of aerial platforms, deliver specific services, and meet demanding performance standards. This shift towards ML-driven control strategies reflects the evolving complexity of NTN communications, encompassing a broader range of non-terrestrial platforms beyond UAVs. These advancements underscore the significance of integrating ML into NTN architectures to enhance efficiency and adaptability in non-terrestrial communication systems [14–16].

Reconfigurable intelligent surfaces (RISs) have recently emerged as a key enabling technology for 6G networks due to their capability to optimize wireless propagation environment, offering extended network coverage in a cost-effective manner. Integrating RISs into HAPs offers significant advantages, particularly in enhancing connectivity. Figure 1 depicts the scenario involving connectivity, where RISs control the phase of the incident signal, using one radio frequency (RF) chain to enhance the signal-to-noise ratio (SNR). As crucial relays between deep space networks (DSNs) and TNs, RISs facilitate efficient communication for interplanetary and Earth-orbiting missions, improving overall communication reliability and efficiency [17,18]. RIS-assisted HAPs provide significant advantages in coverage, connectivity, and energy efficiency. The dense deployment of HAP systems extensively enhances ground coverage, effectively eliminating coverage gaps and ensuring continuous connectivity, especially during handoff processes involving low-Earth-orbit (LEO) satellites [19].



Figure 1. High-altitude platform (HAP) system architecture integrating ARIS for enhanced wireless communication for multiple users [13].

This paper provides a comprehensive survey of RIS-assisted NTN-based 6G communications. The main novelty of our work is highlighted through addressing the following key questions:

- 1. What is RIS technology and how does it integrate with HAP systems?
- 2. Can conventional channel estimation methods and ML-based solutions be applied to RIS and NTN systems?
- 3. How can RISs and NTNs be integrated into 6G technologies?
- 4. How can energy efficiency and sustainability principles be applied to RISs and NTNs?
- 5. What are the future perspectives and challenges faced by RIS and NTN technologies?

The details of the different topics covered in this paper are shown in Figure 2. The remainder of the paper is organized as follows. Section 2 discusses the basic principles of RISs, beamforming optimization techniques for RIS-HAP systems, and applications of

RISs in NTNs. In Section 3, ML-based channel estimation solutions in RIS-HAP systems are elaborated. RIS integration with other communication technologies is presented in Section 4. Section 5 focuses on energy efficiency and sustainability issues in RIS-HAP systems. Section 6 explores the intersection of ML and RISs in wireless communication networks and discusses future perspectives and challenges in implementing RIS-based NTN systems. Finally, Section 7 concludes with a summary of the key findings.



Figure 2. Taxonomy of RIS-assisted NTN-based 6G communications along with blocks illustrating the interconnection of research themes, forming a cohesive framework to accomplish RIS-assisted NTN-based 6G communications.

2. RISs and Their Integration with NTNs

This section navigates the interplay between RISs and HAPs by understanding the core principles that govern the abilities of RISs to manipulate electromagnetic (EM) waves to enhance the performance of communication systems. We first define RISs and compare their superiority to dynamic switching. We then help to differentiate between RISs with traditional beamforming followed by its comparison with other relevant technologies such as relays, massive multiple-input multiple-output (MIMO) systems, etc. Finally, we highlight the advantages of integrating RISs with NTNs and their applications.

2.1. RISs and Their Advantage Against Dynamic Switching

RIS refers to a flat surface comprising several metamaterial reflection components with adjustable phase transitions, enabling manipulation of incoming EM waves' phase shifts. This cutting-edge technology allows for the dynamic control of the wireless environment and identification of reflected EM wave behaviors [20]. Controlling transmitted signals via a reflection array was first introduced in 1997, creating reflection displays using metamaterial elements to manage millimeter wave (mmWave) reflecting surfaces effectively [21,22]. Their work suggests improvements in production, component design, and input layout while demonstrating the beamforming capabilities of reflecting panels in the millimeter wave (mmWave) range. Utilizing RISs can improve communication effectiveness significantly.

Various strategies have been employed in recent years to achieve metamaterial tunability, including embedding switching apparatus externally and integrating positive intrinsic-negative (PIN) diodes with metasurface components [23]. Nowadays, reconfiguration capabilities on metasurfaces have become achievable, and several research teams have developed prototypes. Studies have explored software-controlled passive patch antennas for static beam development, showcasing advancements in beamforming capabilities and reconfiguration in wireless systems [24,25]. Several advancements in RIS technology have been achieved recently, showcasing diverse capabilities and advantages:

- 1. A PIN-based 256-unit RIS with excellent energy density and cost-effectiveness has been developed [25–27].
- 2. Researchers have successfully regulated orientation and transmission reflection using a variable anisotropic computerized sequence alignment metasurface [28]. It was shown that doubling the number of antenna elements from 400 to 800 increased the achievable rate to 2 bps/Hz.
- 3. NTT Docomo from Japan has created a highly transparent two-dimensional metamaterial, allowing for covert usage in exhibitions and advertisements [29].

These advancements highlight the superiority of RIS technology over dynamic switching methods, as RISs offer several advantages, such as low power consumption and simplified interference management. This capability positions RISs as critical enablers for intelligent wireless environments, particularly for 6G [30,31].

2.2. RISs Against Traditional Beamforming Architecture

Compared to traditional beamforming architectures, a significant innovation of this scheme lies in its ability to fully harness the potential of RISs with their unique programmable feature as an external and standalone analog beamformer [32-34]. This eliminates the need for internal analog beamforming at BSs, leading to simplified architecture and a substantial cost reduction. The research discussed in [35–40], addresses the joint beamforming optimization for RIS-aided full-duplex (FD) NTN communication systems. Optimizing beamforming in RIS-aided FD communication systems is essential for enhancing performance across various metrics as it improves signal quality, spectral efficiency, and interference mitigation. In this optimization approach, refining beamforming vectors at both the transmitter and receiver ends and adjusting the phase shifts on RIS elements are jointly carried out. The most important objectives are to maximize the received signal power, minimize interference, and adhere to the communication system constraints, including the handling self-interference inherent in FD setups that share the same frequency spectrum [41–43]. To address these challenges comprehensively, diverse optimization algorithms and techniques such as convex optimization, genetic algorithms, particle swarm optimization, and deep learning (DL) methodologies are deployed. The benefits of RISs are fully harnessed to achieve superior spectral efficiency and interference management by synergizing the optimization of beamforming vectors and phase shifts in RIS-FD systems [44,45]. In [45], the presented framework showed a power reduction of 6 dB at 5 bps/Hz compared to the baseline method. Designing efficient beamforming algorithms considering self-interference and the needs of joint optimization further enhances RIS-aided FD system performance [39,40,46].

The study in [43] focuses explicitly on minimizing total transmit power by combining optimizing active beamforming at the sources and passive beamforming at an RIS positioned between them. The authors showed the effectiveness of RISs for FD schemes, showing that they outperform half duplex when the transmit power is less than 25 dBm. The number of transmit antennas was fixed to 40 and the data rate was set to 4 bps/Hz [43]. They proposed an efficient algorithm such as alternating optimization to solve the nonconvex problem. In addition, joint optimization of beamforming, RIS phase shift, and user association in a multi-cell network aided by multiple RISs were studied [38]. The authors proposed less complex algorithms compared to existing methods and effectively exploited multiple RISs to improve network performance.

The process of joint beamforming optimization for RIS-aided FD communication includes simultaneously optimizing the beamforming vectors at the transmitter and receiver and adjusting the phase shifts of the RIS [47]. It was shown in [47] that the target sum rates of 9.5 and 12.5 bps/Hz can be achieved by saving 58% and 39% of the RIS elements, respectively. This optimization strategy aims to strengthen signal quality at the receiver, boost spectral efficiency, and mitigate interference in RIS-FD systems [48]. By jointly optimizing these parameters, the system can benefit from the RIS's unique advan-

tages, such as enhancing signal strength and directing signals toward intended users or nullifying interference.

2.3. Comparison with Other Technologies

The research in [49] explores an array of different solutions, uncovering significant SNR improvements in massive MIMO and RIS systems. Under far-field conditions, SNR grows linearly with the number of antennas (N) in massive MIMO but quadratically (N^2) with RISs. However, near-field complexities limit RIS's asymptotic SNR despite its faster growth. It emphasizes RIS's potential for enhanced SNR, scalability, and cost-effectiveness compared to massive MIMO. Table 1 gives a comparative overview of these technologies. RIS, massive MIMO, relays, and backscatter operate across several parameters such as duplex capability, the number of transmit RF chains, hardware cost, and energy consumption. RISs, operating through passive or active reflection, stand out with their FD capabilities and zero RF chain requirement, leading to low hardware costs and minimal energy consumption. In contrast, although offering full or half duplex options, massive MIMO and relay systems necessitate numerous RF chains, resulting in significantly higher hardware costs and energy usage. Backscatter, similar to the RIS in terms of low cost and energy needs, operates passively but generally underperforms compared to RIS.

| Attribute | RIS | Massive MIMO | Relay | Backscatter |
|------------------------|-------------------------------|---------------------------------------|--|---------------------|
| Operating Mechanism | Passive/Active, reflection | Active, transmission/ reception | Active, reception and transmission | Passive, reflection |
| Duplex | Full | Half/full | Half/full | Full |
| No. of RF Chains | 0 | Ν | Ν | 0 |
| Hardware Cost | Low | Very high | High | Very low |
| Energy Consumption | Low | Very high | High | Very low |

Table 1. Comparative analysis of various technologies with the RIS.

2.4. Recent Advances in RIS-Integrated NTNs: Unveiling Progress

Recent advances in RIS-integrated NTNs highlight resource optimization and performance analysis developments. Researchers are delving into strategies for power consumption minimization, energy efficiency enhancements, and sum-rate maximization, utilizing optimization techniques. These studies span from exploring joint beamforming strategies to designing optimal phase shifts, showcasing the adaptability and efficiency of RISs across various scenarios [50]. Researchers actively explore secure cooperative communications and employ alternating optimization strategies in RIS-integrated NTNs. These multifaceted contributions extend to comprehensive performance analysis, with studies deriving expressions for outage probability, symbol error rate, ergodic capacity, secrecy, and connection probability [51,52]. This collective body of work illuminates the versatility and robustness of RISs in enhancing the efficiency, security, and overall performance of NTN landscapes [35,53,54]. The simulation results in [7] revealed that our proposed robust scheme significantly outperforms the RIS with random phase, the non-robust scheme, and the secure transmission scheme without the RIS, achieving improvements of approximately 8%, 17%, and 41%, respectively, in terms of friendly interference performance. From addressing fundamental link challenges to enhancing security and performance, RIS stands as a cornerstone in shaping the future of NTNs.

We critically examined beamforming methods customized for HAP systems equipped with RIS technology. Some of the core advantages found are as follows:

- Enhanced Network Coverage: In scenarios where direct line-of-sight (LoS) links are blocked or absent, RISs can create virtual LoS links between ground users and HAPs in air–ground networks, significantly improving communication quality and extending network coverage [8,55–57].
- **Dynamic and Smart Control:** Mounting the RISs on a HAP allows for the dynamic control of the RIS's location, leveraging the HAP's mobility. This mobility introduces a new degree of freedom in the RIS's design compared to terrestrial RISs (TRISs) installed at fixed locations as shown in Figure 1 [58,59].
- **Expanded Wireless Coverage:** When installed on an HAP, the RIS will likely establish an LoS path for transmitting and receiving components. Unlike TRISs, which support communication between two nodes in half the space, an RIS on an HAP can expand wireless coverage and achieve full-angle reflection [60].
- Ultra-wide Bandwidth in THz or mmWave Band: An HAP and RIS can operate in the terahertz (THz) or mmWave band, offering an ultra-wide bandwidth to meet high-data-rate requirements [61]. The mobility of HAPs and reconfigurability of RISs efficiently compensate for the high path loss and blockages typical in THz or mmWave links [60].
- Enhancing Communication Links: An imperative motive for incorporating RISs into NTNs is their ability to surmount the intricate challenge of closing communication links. Designed explicitly for scenarios involving mobile ground users with lowdirectional antennas, RISs strategically compensate for path losses. This intervention significantly enhances signal reception, resulting in elevated data rates and an overall boost in system performance [59].
- **Physical-Layer Security and Reliability:** HAP communication often suffers from LoS blockage and vulnerability to eavesdropping. An RIS can mitigate external interference, enhancing physical-layer security and reliability. For instance, passive beamforming through an RIS directs beams away from eavesdroppers while strengthening signals for intended users. This approach improves security and reliability in HAP communication, primarily when airborne RISs (ARISs) support performance enhancement after optimizing HAP trajectory or location [13,59,62,63].
- Enhanced Data Rate and SNR: RIS-assisted HAP communication aims to boost the data rate and SNR. Factors such as the number of RIS-reflecting elements, RIS location, and HAP-hovering position are considered. Optimization approaches are adopted to enhance the received signal power at the HAP [16,60,64].
- **Mitigating Interference and Spectrum Reuse:** Coexisting with TNs brings forth the challenge of co-channel interference in NTNs. An RIS emerges as a strategic tool for interference mitigation, creating nulls or constructive interference in specific directions [65,66]. This proactive interference management minimizes disruptions, ensuring efficient spectrum reuse and safeguarding the integrity of both TNs and NTNs.
- Energy Efficiency and Resilience: Energy efficiency takes center stage in NTNs, impacting mission duration, costs, and sustainability. Operating passively without active transmissions, RISs become a key player in enhancing energy efficiency and fostering green communications [65,66]. Moreover, RISs contribute to NTN resilience by providing an additional layer of redundancy. RISs adeptly redirect signals in the face of obstacles, atmospheric conditions, or disruptions to terrestrial networks, ensuring continuous and reliable communications.

This comprehensive examination illuminates the multifaceted contributions of RISs in shaping the future of non-terrestrial networks and their pivotal role in next-generation wireless communication systems [67,68].

2.5. Application of RISs to NTN Communications

Incorporating RISs into NTNs and upcoming 6G technologies presents distinct advantages. RIS's ability to manipulate EM waves without conventional transceivers aligns well with the objectives of NTNs, particularly in reducing payload weight and power consumption, crucial factors in satellite operations [69]. For 6G, RISs can be pivotal in realizing ultra-reliable, low-latency communication and enhanced connectivity. Its energy-efficient nature and reduced hardware requirements offer an economical and sustainable solution for the expansive infrastructure envisioned in 6G. The unique properties of RISs, including improved signal propagation and coverage, address the challenges in NTNs and 6G, such as large coverage areas and signal attenuation, making it an innovative and viable technology for 6G networks.

2.5.1. LEO Satellites

In 6G, LEO satellites may provide connectivity in remote and underserved areas. The LEO satellite has drawn the most attention these days among the various kinds of satellites working at varied altitudes in orbit because of its low height as well as the benefits that come with it, including less signal attenuation, a shortened transmission time, and a reduced expense in comparison to other satellites on a higher altitude [30,70,71]. The system model of an RIS-assisted HAP is illustrated in Figure 3, which includes an energy-constrained HAP with *M* antennas, an RIS equipped with *N* reflecting elements, and *K* single-antenna users [72]. The LEO satellite's transmission bandwidth is still significantly greater than ground transmission, resulting in orders-of-magnitude of more significant route degradation. As a result, high-power broadcasters and high-sensitive receptors are typically used in LEO satellite technology to make up for the high route loss [73]. Nevertheless, doing so inevitably raises the expense and energy usage of the gear. As a substitute, the conventional passive reflect array has been used in satellite/deep-space telecommunications to accomplish unidirectional broadcasting for route loss mitigation without appreciably raising the equipment cost and energy usage [73–77].



Figure 3. RIS-assisted HAP system model facilitating communication between a LEO satellite and a user via dual signal paths.

2.5.2. Microelectromechanical Systems and Metasurfaces

RISs or their various counterparts are the result of recent advancements in technology in microelectromechanical systems (MEMSs) and metasurfaces. These technologies have made it possible to repurpose passive reflections in real-time, significantly adjusting the inert transmission and reflection [78]. This development enables the creation of efficient, cost-effective wireless communication streams. Generally, RIS is a large EM metasurface composed of numerous, inexpensive static reflecting components. Each element can independently control the phase shift and frequency of incident signal, significantly boosting the power of the reflected signal in a desired direction, enabling fine-grained passive beamforming gains [79]. RISs are capable of handling the time-varying wireless channels of high-mobility LEO satellites and of compensating for the significant path loss caused by vast transmission distances thanks to its real-time tunability [80,81]. Moreover, unlike traditional active beamforming systems for satellite technology, RISs offer full-duplex, unobtrusive beamforming at minimal cost and power consumption, without the need for active transmitters or RF networks for signal processing or amplification [82].

2.5.3. RIS-Assisted Communication Networks

Numerous studies on using RISs in different communication networks, including non-orthogonal multiple access (NOMA) and orthogonal frequency-division multiplexing (OFDM) systems, have been inspired by the RIS's attractive characteristics [83–89]. The fundamental idea of RISs, where the RIS is installed on the satellite or close to the terrestrial users to improve their network throughput, has just been applied to satellite relay in some earlier research [90,91]. These works, nevertheless, assumed the ideal channel estimation for the design and operation of detached beamforming and neglected to take into account the crucial channel approximation/beam traceability problem for RISs, which remains a difficult task in practice owing to its numerous reflecting components that do not have data communication or processing capabilities, particularly for the scenario of LEO satellites with excellent mobility [92–94]. Even though effective passive beamforming models and channel coding strategies have been developed for various RIS-supported structures, most of these are focused on terrestrial communication networks with BSs and RISs fixed in position to serve steady-state users, which might not be suitable for high-mobility LEO satellite systems [95]. For example, the fast mobility of LEO satellites might lead to highly time-varying wireless channels with ground nodes (GNs) and satellites which can complicate regular channel estimation and beam management for both active and inactive beam positioning. Such challenges could significantly degrade communication performance [95,96]. Furthermore, the route identification and beam monitoring in RISsupported LEO satellite communications are more difficult owing to the substantial Doppler impact generated by the LEO satellite's rapid motion [96,97].

3. Conventional Channel Estimation and Machine Learning-Based Solutions

Classical optimization algorithms face significant challenges in meeting the demands of emerging applications in RIS-assisted integrated satellite (RIS-IS) operations and NTNs. These challenges primarily arise from the complexity of communication scenarios, unknown channel models, and difficulties in accurately modeling dynamic environments through mathematical expressions, which is further exacerbated by non-linearities due to hardware losses. The continuously changing configuration of RISs further complicates matters, hindering traditional methods' applicability for real-time operations such as Internet of Things (IoT) or Internet of Vehicles (IoV) environments.

However, deep reinforcement learning (DRL), a combination of DL and reinforcement learning (RL), shows its effectiveness in handling these challenges. DRL's ability to operate with minimal labeled training data makes it adaptable for optimizing wireless communication systems. This is especially true for complex and dynamic channel environments. Without total channel model information, DRL constructs wireless channel knowledge by interacting with the environment and receiving rewards. When combined with efficiently designed algorithms, sequentially finding optimal solutions to multi-objective optimization problems becomes a reality when this neural network is combined with efficiently designed algorithms. This approach excels in joint optimization-based networks, such as beamforming design, power allocation strategy, interference coordination, and user distribution model tracking in diverse communication scenarios, showcasing its usefulness in addressing the delicate challenges of RIS-assisted NTNs. With its ability to navigate complex and dynamic channel environments using minimal labeled training data, DRL showcases its superiority over standard schemes. The comparison emphasizes DRL's efficiency in real-time decision-making for joint optimization tasks, such as transmit beamforming and RIS phase shift matrix optimization. The adaptability and effectiveness of

DRL in addressing the intricate challenges of RIS-assisted NTNs stand out from traditional methods, suggesting that it represents a more advanced and efficient approach to channel estimation in this context.

A direct examination of the reciprocity in channel estimation for both uplink and downlink communication with RISs reveals fascinating dimensions rooted in actual scientific understanding [98]. While fully acknowledging the established reciprocity in time division duplexing, introducing RISs as something that breaks the reciprocity introduces genuine intrigue. Research on AI and ML algorithms could be pivotal in mapping channel responses and optimizing RIS behavior based on incident angles and frequency responses, presenting a solid approach to tailoring specific deployment scenarios [98,99]. This fusion of AI, ML, and RIS optimization gives the potential for advancements in wireless communication systems, particularly when considering the actual science behind optimizing phase shifts and channel state information (CSI) in NTN environments [98–100].

Integrating AI and ML to realize RISs for communication in HAPs and NTNs presents realistic possibilities based on ongoing research [8]. The article emphasizes the practical application of DL to navigate challenges in modeling the propagation environment. Reducing the dimensions of channel estimation and discerning deployment-specific characteristics can enhance the accuracy of the learning algorithms in local propagation conditions and direct optimal signal reflection on expansive surfaces. This aligns with the complex realities of user locations and the intricate effects of mutual coupling that prove to be cumbersome for traditional modeling techniques [84].

Moreover, integrating AI and ML introduces a functional solution to configure largescale RIS deployments. This underscores the potential for learning algorithms to bypass the necessity for precise physical modeling, offering efficient solutions in resolving optimization issues related to configuring surfaces with numerous elements [101]. This suggests that AI and ML can streamline the design and operation of RISs, especially in scenarios where conventional mathematical approaches might not be feasible or are time-consuming [101].

The best ML algorithm for optimizing RISs in NTNs, particularly for tasks like optimizing phase shift and CSI, would likely be reinforcement learning, specifically Deep Q-Networks (DQNs). DQNs excel in scenarios where the system needs to learn optimal actions in dynamic and changing environments [102]. In the context of RIS-aided NTNs, where adaptation to real-time conditions is crucial, a DQN's ability to handle complex decision-making processes and learn from experience makes it a robust choice. The use of AI or ML further gives the capability to navigate the challenges posed by mutual coupling effects and dynamically adjust RIS configurations to align well with the requirements of optimizing phase shift and CSI in an NTN setting.

4. Integration with Communication Technologies

4.1. Synergy Between RISs and Current, Future Wireless Communication Systems Including 6G

Integrating RISs into existing and forthcoming communication systems, including the anticipated 6G, signifies a transformative leap towards more dynamic, efficient, and robust communication networks [103,104]. The RIS holds immense potential in synergizing with and enhancing current and future communication protocols [105]. This section explores how RISs can be harmoniously integrated with current and future wireless communication systems, mainly focusing on the evolution towards 6G.

1. Enhancing 5G Networks and Beyond: As the world gradually transitions from 5G to 6G, RISs are poised to play a crucial role in addressing the limitations of previous generations and enhancing network performance [104]. Fifth-generation networks, characterized by high data rates, low latency, and massive device connectivity, can benefit from RIS's ability to provide surfaces that improve signal propagation and coverage, especially in urban environments where obstacles are prevalent [103]. RISs can mitigate signal degradation and enhance the efficiency of massive MIMO systems, which are integral to 5G technology [106,107].

- 2. **Empowering** 6G Vision with RISs: The envisioned 6G networks aspire to provide ubiquitous wireless intelligence, extreme connectivity, and profound integration with AI. RISs inherently align with these goals, offering adaptive environments that can be optimized for various communication needs. By controlling the propagation of electromagnetic waves, RISs can help 6G to achieve its targets of THz communications, extremely low latencies, and seamless integration of terrestrial, aerial, and space networks [103,105].
- 3. **Synergy with IoT and Machine-Type Communications:** The IoT and Machine-Type Communications (MTCs) are expected to increase under 6G, with billions of devices communicating simultaneously [105]. RISs can enhance these devices' energy efficiency and reliability by optimizing signal paths and reducing interference, thereby extending the battery life and improving the performance of IoT and MTC networks [108,109]. This is especially crucial in dense networks where traditional methods of ensuring quality of service may fall short.
- 4. **Facilitating Advanced Beamforming and Spatial Diversity:** Advanced beamforming and exploiting spatial diversity are critical techniques in modern communication systems to improve signal quality and system capacity. RISs can provide a cost-effective means to implement these techniques, offering finer control over the signal propagation environment [110]. By dynamically adjusting the phase shifts of individual elements, RISs can create constructive or destructive interference where needed, significantly enhancing the capabilities of beamforming and MIMO technologies in current and future communication systems [111].
- 5. Integrating with Software-Defined Networks (SDNs) and Network Function Virtualization (NFV): As networks become more software-driven, integrating RISs with SDNs and NFV can lead to highly flexible and adaptable network architectures. RISs can provide physical-layer adaptability, while SDNs and NFV can dynamically adjust network resources and functions [105]. This synergy optimizes real-time network performance, meeting modern communication systems' diverse and dynamic requirements.
- 6. Enabling Energy-Efficient and Sustainable Networks: As global communication demands surge, so does the energy consumption of communication networks. RISs offer a pathway to more sustainable networks by enhancing signal efficiency and reducing the need for power-intensive transmitters and infrastructure [111]. This aligns with the green communication objectives of 6G, which emphasize sustainability alongside technological advancement.
- 7. Security Enhancements: Security is a paramount concern in modern communication systems. RISs can contribute to more secure communications by selectively enhancing the signal strength in the intended directions while mitigating eavesdropping risks. Moreover, the dynamic nature of RISs can provide an additional layer of security through variable and unpredictable propagation environments, complicating potential interception or jamming attempts [110,112].

The synergy between RISs and current and future communication systems, including the journey towards 6G, is a testament to the ongoing evolution of wireless technology. The RIS is a pivotal technology that enhances signal propagation, system capacity, energy efficiency, and security [112]. It aligns with the ambitious goals of future communication systems, promising to address the burgeoning demand for high-quality, ubiquitous, and sustainable connectivity. As research and development continue to unfold the potentials of RISs, their integration with current and future communication systems is likely to become increasingly central, paving the way for an era of adaptive, controlled wireless communication environments.

4.2. RIS's Potential in Enhancing Radar and Navigation Systems

Including RISs in radar and navigation systems represents a significant technological advancement, promising enhanced performance and new functionalities [110,113,114]. RIS,

essentially a surface with the ability to alter EM waves passing through or reflecting off it, can dramatically improve the efficacy and capabilities of radar and navigation systems in several ways.

- 1. **Signal Enhancement and Path Loss Reduction:** RISs can reflect signals toward specific directions, mitigating path loss and enhancing signal strength at the receiver. RISs can provide alternative non-LOS paths in complex urban environments or terrains where LOS is obstructed, significantly improving coverage and reliability [110].
- 2. **Target Resolution and Detection:** RISs can contribute to a higher resolution and better target detection in radar systems. By controlling the phase and amplitude of the reflected waves, RISs can help to distinguish between closely spaced objects, enhancing the radar's ability to detect and track objects with improved accuracy [110,115].
- 3. **Reducing Interference and Clutter:** Unwanted signals can be mitigated when an RIS is set up correctly. It will reduce interference and improve the SNR [113]. This is particularly helpful in crowded frequency bands such as those used by frequency-modulated continuous-wave (FM-CW) radar or navigation signals, which are prone to interference from multiple sources. By focusing the signal in desired directions and avoiding undesired ones, RISs can significantly reduce clutter and enhance the system's performance.
- 4. **Dynamic Beamforming and Steering:** Traditional radar and navigation systems require mechanical steering of antennas or complex phased array systems for beamforming. RISs can provide a more efficient and faster means to control the wavefront, enabling dynamic beamforming and steering without moving parts or expensive arrays [113]. This can lead to systems that are more responsive, precise, and less prone to wear and tear.
- 5. Enhanced Multipath Exploitation: RISs can be strategically leveraged to harness the dynamic multipath propagation environment; a scenario unequivocally deemed detrimental in conventional systems [110,116]. By meticulously controlling the properties of the surfaces within the environment, RISs can transform the multipath effects from a formidable challenge into a potent advantage, utilizing these phenomena to reinforce system robustness and glean unparalleled insights about the communication environment.
- 6. **Improved Security:** The ability of an RIS to control electromagnetic waves can also be leveraged to improve the security of multi-static radar and localization signals [104,105]. By ensuring that signals are only strong in intended directions and locations, RISs can make it more difficult for adversaries to intercept or jam the signals in an electronic warfare situation, providing a layer of security that is actively adaptable to threats.

Hence, integrating RISs into radar and navigation systems holds substantial potential, from enhancing signal quality and system performance to opening up new capabilities and improving security. As research intensifies and practical implementations materialize in this field, we may be on the brink of a new era in radar and navigation technology, marked by unprecedented adaptability, enhanced efficiency, and improved robustness. The capabilities of RISs in the beyond 5G (B5G) networks is just beginning to be explored, and the coming years will likely see significant advancements and innovative applications of this promising technology.

5. Energy Efficiency and Sustainability

Integrating RISs with HAPs introduces a groundbreaking approach that enhances communication capabilities and addresses some critical energy efficiency and sustainability concerns. The demand for wireless communication has surged, driving the quest for improved communication system performance, especially in remote areas. RIS-enhanced HAPs is among the technologies that offer promising energy efficiency and sustainability solutions in wireless networks.

5.1. Energy Efficiency in the Context of RIS and HAP Integration

HAPs are used for various purposes, including telecommunications, surveillance, and environmental monitoring. However, the challenge lies in the energy constraints associated with sustaining operations at such altitudes. Traditional communication systems on HAPs often face limitations in terms of energy consumption, impacting their endurance and overall effectiveness. When equipped with RISs, these platforms can significantly improve their performance metrics, particularly regarding energy efficiency. Energy efficiency is critical to RIS-enhanced HAPs, and researchers have devoted considerable efforts to analyzing its impact [72,102]. They have studied the energy efficiency of RIS-enhanced HAPs in various scenarios, such as multiple-input single-output (MISO) and MIMO networks. RIS technology significantly reduces energy consumption by optimizing various parameters such as time allocation, HAP energy beamforming, receiving beamforming, user transmit power allocation, and the RIS energy reflection coefficient [65]. Analyzing the energy efficiency of RIS-enhanced HAPs involves various factors, such as transmit power, channel conditions, data rates, and the energy consumption of the RIS itself, which must be considered [117]. RIS placement, configuration, and optimization techniques also affect energy efficiency [118].

In the context of RIS-enhanced HAPs, the multi-objective optimization approach involves the joint optimization of various parameters. This includes time allocation, HAP energy beamforming, receiving beamforming, user transmit power allocation, and the energy reflection coefficient of the RIS. This comprehensive optimization scheme aims at maximizing energy efficiency while ensuring fair and equitable distribution of resources among users in the network [119]. The optimization framework offers a significant step towards realizing the full potential of RIS-enhanced HAPs in meeting the ever-increasing demands for wireless communication while fostering sustainability and fairness in wireless networks. Using RIS reduces the number of reflections in the channel, resulting in a favorable channel that requires less transmit power for a given data rate [120].

Optimizing the multi-user transmit beamforming power requirement for the active user set, researchers have proposed computationally efficient algorithms to minimize network power consumption and improve energy efficiency [121]. Moreover, introducing RIS-aided networks has enhanced energy efficiency [122]. In RIS-aided networks, the joint optimization of active beamforming and phase shifter design has significantly improved system performance and energy efficiency [123]. To address the energy efficiency of RIS-enhanced HAPs, various algorithms [65] and optimization techniques [124] have been developed. One approach is optimizing resource allocation by formulating a convex optimization problem and proposing low-complexity sub-optimal algorithms. Optimizing the BS's precoding matrix and the reflecting surface's reflection coefficient has maximized the weighted minimum rate for all users to ensure network fairness and minimize energy consumption [65].

5.1.1. Energy Efficiency Through Adaptive Beamforming

One of the primary advantages of incorporating RISs into HAPs is the ability to adapt to the propagation environment dynamically. RISs can effectively modify the phase and amplitude of the incoming EM waves to create constructive interference in desired directions [125]. This beamforming capability means that signals can be focused more accurately on specific receivers, significantly reducing the power required to maintain a reliable communication link, thus enhancing the overall energy efficiency of the HAPs [72,116,125].

5.1.2. Extending Coverage with Reduced Power

Traditional HAPs require significant energy to increase their coverage area, often increasing power consumption [72]. With RISs, HAPs can extend their coverage by intelligently reflecting signals toward blind spots or areas beyond their direct line of sight without additional power expenditure. This ability to manipulate the signal path passively means that more areas can be covered with the same or even less energy, improving the energy efficiency ratio [72,125].

5.1.3. Minimizing Losses Through Passive Elements

RIS consists of numerous low-cost, passive elements that do not require power for signal processing or active radiation. This passive nature significantly reduces the system's energy consumption [72,116]. Unlike active antenna systems, where each element might require its power source for operation, RISs can manipulate the EM waves impinging upon them without any active energy input, leading to substantial energy savings [126].

5.1.4. Reducing Interference and Enhancing Signal Quality

By controlling the propagation of signals, RISs can also minimize interference, a common problem that requires additional energy expenditure in traditional systems to overcome [103,105]. RISs can improve the SNR by directing signals away from unintended receivers and toward the intended targets. This means that signals require less power to achieve the same level of quality and reliability, further enhancing energy efficiency.

5.1.5. Operational Longevity and Maintenance

RIS's passive nature and lack of moving parts also mean less wear and tear, leading to longer operational life and reduced maintenance needs. This aspect indirectly contributes to energy efficiency by reducing the energy and resources required for manufacturing, deploying, and maintaining the HAPs [72,126].

5.1.6. Enabling Green Communication Networks

Integrating RISs into HAPs aligns with the broader objective of developing sustainable and green communication networks [126]. By reducing the energy requirements of HAPs, RISs contribute to lowering the overall carbon footprint of communication infrastructure. This is particularly pertinent as the demand for data and connectivity continues to grow exponentially, necessitating solutions that can scale without proportionally increasing energy consumption.

Analyzing the energy efficiency of RIS-enhanced HAPs reveals their potential to optimize power consumption and improve overall system performance [127]. The strategic deployment of RIS technology can enhance the coverage and capacity of wireless networks while reducing the energy consumption of BSs [124]. Hence, the energy efficiency of RIS-enhanced HAPs is multifaceted, stemming from the ability to direct and manipulate signals efficiently, extend coverage with minimal energy increase, and reduce interference. The passive nature of RISs contributes significantly to lowering operational energy demands, making it a promising technology for sustainable and efficient high-altitude communication platforms. As technological advancements continue to unlock the potential of RISs, it is expected that HAPs will become even more energy-efficient, paving the way for an era of more sustainable communication and monitoring solutions in the stratosphere.

5.2. Sustainable Practices in Deploying RIS Technology

Sustainable practices are crucial in deploying RIS technology in HAPs and are paramount to ensure long-term environmental benefits. These practices include energy-efficient hardware design, optimized power management algorithms, integration of energy-harvesting technologies [67], and the use of renewable energy sources for HAPs, such as solar or wind energy [127]. The deployment of RIS technology also involves considering the environmental impact and resource management [128]. This includes minimizing electromagnetic pollution and reducing the exposure of the end user to electromagnetic fields [129]. Moreover, energy-efficient algorithms and optimization techniques can be applied to maximize the energy harvested by RIS-enhanced HAPs, leading to a more sustainable and eco-friendly operation [130].

Various approaches have been recently explored to ensure the long-term sustainability of RIS-enhanced HAPs. Factors such as energy consumption and resource allocation optimization have been considered [124]. These practices include optimizing power allocation, beamforming strategies, and integrating renewable energy sources into the HAP's infrastructure. For example, a novel mechanism for a new element on–off to improve the scalability and energy efficiency of RIS-enhanced HAPs is presented in [122]. This mechanism allows RIS elements to selectively activate or deactivate based on channel conditions and user requirements, reducing unnecessary energy consumption and improving overall system sustainability. Furthermore, implementing energy-saving mechanisms, such as sleep mode and dynamic power control, can further enhance energy efficiency in RIS-enhanced HAPs.

Developing mitigation strategies to tackle the environmental impact based on assessment must be performed to identify potential risks. These strategies include considering the materials used in manufacturing RIS elements, ensuring proper disposal and recycling processes, and minimizing the overall carbon footprint of RIS-enhanced HAPs [128].

Deploying RIS technology in HAPs will foster a more eco-friendly and energyefficient wireless communication system. This can be achieved through several approaches, such as follows:

- Community pricing: Rather than individual householders bearing the cost of RIS deployment, community developers can price the deployment of RISs [131]. This approach reduces complexity and promotes collective responsibility in implementing RIS technology.
- Function pricing: Customizing the deployment and size of RISs based on different communication needs can optimize resource allocation and promote efficient usage of RIS technology.
- Renewable energy integration: Integrating renewable energy sources, such as solar or wind power, into RIS deployment can reduce reliance on traditional energy sources and promote sustainability by reducing greenhouse gas emissions.
- Lifecycle assessment: Conducting a thorough lifecycle assessment of RIS technology, including its manufacturing, deployment, maintenance, and disposal, helps to identify areas where environmental impact can be minimized and sustainability practices can be improved.
- Standardization: Developing industry-wide standards for RIS technology ensures interoperability and compatibility, reducing unnecessary duplication of resources and promoting efficient deployment practices.
- Recycling and reuse: Implementing processes for recycling and reusing materials used in RIS technology, such as electronic components, can reduce waste and promote sustainability.
- Energy-efficient infrastructure: Designing RIS systems with energy-efficient components and technologies, such as low-power-consumption RIS types, can reduce the RIS technology's overall energy consumption and environmental impact.

6. ML-Driven RIS-NTN Systems: Future Perspectives and Challenges

The RIS is a groundbreaking technology in wireless communication networks, potentially significantly changing how data are transmitted, received, and handled [132]. Through the intelligent reflection and manipulation of EM waves, RISs can improve signal strength, reduce interference, and enhance network performance across various settings [120,133]. However, conventional optimization strategies relying on CSI encounter intrinsic constraints that limit their effectiveness in fully utilizing the potential of RIS technology. These limitations arise from wireless channels' intricate and fluctuating nature and the computational burden linked to real-time adjustments and optimization. Therefore, a critical requirement is for more effective and adaptable optimization approaches to leverage the complete capabilities of RIS-enhanced networks [15,134,135].

Recently, ML has become a promising method for overcoming the optimization obstacles that RIS technology encounters. Through utilizing data-driven observations and adaptable algorithms, ML methods can enhance RIS performance in real-time by constantly adjusting phase shifts and beamforming patterns to maximize signal strength and reduce interference [136,137]. Advancements in optimization driven by ML have shown considerable enhancements in RIS performance in various areas, such as mapping, propagation modeling, localization [138], signal processing, and resource allocation. ML algorithms can dynamically

analyzing large datasets and identifying intricate wireless channel behavior [126,135]. Nevertheless, while ML-driven optimization holds potential, various challenges and research hurdles must be tackled. One notable difficulty involves the requirement for extensive datasets to train ML models. Moreover, real-time optimization presents significant computational complexity, and ensuring the resilience of ML algorithms in dynamic and uncertain wireless environments is also a significant obstacle [15,126]. Furthermore, integrating ML with RIS technology necessitates addressing interoperability issues and guaranteeing smooth integration with current network architectures.

optimize RIS setups to accommodate evolving environmental conditions and user needs by

Looking to the future, the combination of ML and RIS technology has the potential to transform wireless communication networks, leading to more effective, adaptable, and intelligent systems. By tackling the issues associated with conventional optimization methods and leveraging data-driven insights, ML-driven RIS optimization offers groundbreaking improvements in network performance, dependability, and scalability. The convergence of ML and RIS marks a significant milestone in advancing wireless communication networks. Through the utilization of adaptive optimization and intelligent reflection principles, this ML-driven RIS technology offers an opportunity to unlock the complete capabilities of future wireless networks, thereby influencing connectivity in the digital era [16,134,139].

Several future challenges for implementing RISs in NTNs using ML techniques are presented as follows:

Resource-efficient RL Methods: Despite promising results, one significant challenge is developing accurate RL methods with moderate computational and energy demands suitable for aerial platforms. The focus should be on ensuring that RL techniques comply with the resources available to the aerial platforms, especially considering real-world scenarios' dynamic and stochastic nature.

Integration with Advanced Technologies: The evolution of NTNs to support superhigh-definition (SHD) and extremely high-definition (EHD) videos with super-high throughput demands and ultra-reliable low-latency communications presents a complex challenge. Integrating RISs into this landscape becomes a multifaceted task. ML methods must adapt to efficiently support 3D virtual reality, space and underwater connectivity, autonomous devices, backscatter communication, and energy harvesting, further adding to the intricacy of the problem.

Discrete Phase Shift Optimization: The emergence of discrete phase shift models due to hardware limitations poses a significant challenge. Current techniques for solving discrete phase shift problems, such as quantization or brute-force searching, have performance degradation or complexity limitations. The need for advancements in optimization algorithms or adapting DRL algorithms to provide more efficient solutions is pressing, as more than the current solutions are needed to meet the demands of the problem.

While RL shows promise for attaining optimal control strategies in wireless communications, there is a need to create precise RL techniques that are both computationally efficient and energy-efficient, particularly for aerial systems. These methods, once perfected, can accurately simulate real-world multi-agent uncertain situations and seamlessly interface with 3D virtual reality to fulfill requirements for SHD, EHD, high-throughput, and ultra-reliable low-latency communications, offering a bright future for this field. The concurrent optimization of active and passive beamforming poses significant challenges, particularly when dynamically sensing channel state information in NTNs with RIS assistance. New RL architectures such as soft-deep deterministic policy gradient (DDPG) algorithms have been proposed to address these challenges, but further investigation is needed to improve network performance in real-time and adapt to the evolving variations in RIS reflective element amplitudes [140,141].

Developing simplistic methods that can deliver results comparable to exhaustive search techniques continues to present a significant challenge. While traditional op-

timization strategies such as the greedy algorithm hold potential, it is crucial to enhance the efficiency of RL algorithms in wireless applications in order to address these challenges effectively.

7. Conclusions

In this paper, we have summarized the latest research activities on the emerging field of RIS-assisted NTN-based 6G communications. By addressing critical user needs such as enhanced security, energy efficiency, expanded range, and high data rates, the integration of RISs has been demonstrated as a key enabler of future wireless communication systems. The synergy between RISs and MIMO systems, as well as advanced radio communications, highlights the importance of AI and ML in optimizing RIS-based beamforming. These technologies work together to solve pressing scientific and engineering challenges while promoting sustainability in NTN operations. The findings in this paper emphasize the pivotal role RISs will play in unlocking the full potential of NTNs, driving the evolution of next-generation wireless technologies. By shedding light on the benefits of RISs for 6G NTNs and proposing future research directions, this work offers a comprehensive understanding of how RISs can revolutionize the field and catalyze the development of highly efficient and innovative wireless communication systems.

Author Contributions: Conceptualization, C.E.W., F.A.K., Q.Z.A. and T.A.; Methodology, C.E.W., F.A.K. and Q.Z.A.; Software, C.E.W.; Validation, C.E.W.; Formal analysis, C.E.W.; Investigation, C.E.W.; Writing—original draft, C.E.W., F.A.K., Q.Z.A., P.S. and T.A.; Writing—review and editing, P.S. and T.A.; Supervision, F.A.K. and Q.Z.A.; Funding acquisition, P.S. and T.A. All authors have read and agreed to the published version of the manuscript.

Funding: Part of this work is supported by the European Union through the Horizon Europe Research and Innovation Programme under the Marie Skłodowska-Curie grant agreement No. 101086218.

Institutional Review Board Statement: Not applicable.

Informed Consent Statement: Not applicable.

Data Availability Statement: Data are contained within the article.

Conflicts of Interest: The authors declare no conflicts of interest.

Abbreviations

The following abbreviations are used in this manuscript:

| Sixth Generation | 6G |
|------------------------------------|------|
| Artificial Intelligence | AI |
| Base Station | BS |
| Deep Deterministic Policy Gradient | DDPG |
| Deep Learning | DL |
| Deep Q-Network | DQN |
| Deep Reinforcement Learning | DRL |
| Deep Space Network | DSN |
| Electromagnetic | EM |
| Extremely High Definition | EHD |
| Full-Duplex | FD |
| High-Altitude Platform | HAP |
| Integrated Satellite | IS |
| Internet of Things | IoT |
| Internet of Vehicles | IoV |
| Line of Sight | LoS |
| Low Earth Orbit | LEO |
| Machine Learning | ML |

| Machine-Type Communication | MTC |
|--|------|
| Markov Decision Process | MDP |
| Microelectromechanical System | MEMS |
| Multiple-Input and Multiple-Output | MIMO |
| Network Function Virtualization | NFV |
| Non-Orthogonal Mutual Authentication | NOMA |
| Non-Terrestrial Network | NTN |
| Orthogonal Frequency-Division Multiplexing | OFDM |
| Positive Intrinsic Negative | PIN |
| Radio Frequency | RF |
| Reconfigurable Intelligent Surface | RIS |
| Reinforcement Learning | RL |
| Signal-to-Noise Ratio | SNR |
| Software-Defined Network | SDN |
| Super High Definition | SHD |
| Terahertz | THz |
| Terrestrial RIS | TRIS |
| Terrestrial Network | TN |
| Unmanned Aerial Vehicle | |
| | |

References

- Zeng, Y.; Wu, Q.; Zhang, R. Accessing From the Sky: A Tutorial on UAV Communications for 5G and Beyond. *Proc. IEEE* 2019, 107, 2327–2375. [CrossRef]
- Zeng, Y.; Zhang, R.; Lim, T.J. Wireless communications with unmanned aerial vehicles: Opportunities and challenges. *IEEE Commun. Mag.* 2016, 54, 36–42. [CrossRef]
- Motlagh, N.H.; Bagaa, M.; Taleb, T. UAV-Based IoT Platform: A Crowd Surveillance Use Case. *IEEE Commun. Mag.* 2017, 55, 128–134. [CrossRef]
- 4. Abbas, W.B.; Ahmed, Q.Z.; Khan, F.A.; Mian, N.S.; Lazaridis, P.I.; Sureephong, P. Designing Future Wireless Networks (FWN)s with Net Zero (NZ) and Zero Touch (ZT) Perspective. *IEEE Access* 2023, *11*, 83301–83321. [CrossRef]
- Chandrasekharan, S.; Gomez, K.; Al-Hourani, A.; Kandeepan, S.; Rasheed, T.; Goratti, L.; Reynaud, L.; Grace, D.; Bucaille, I.; Wirth, T.; et al. Designing and implementing future aerial communication networks. *IEEE Commun. Mag.* 2016, 54, 26–34. [CrossRef]
- Naous, T.; Itani, M.; Awad, M.; Sharafeddine, S. Reinforcement Learning in the Sky: A Survey on Enabling Intelligence in NTN-Based Communications. *IEEE Access* 2023, 11, 19941–19968. [CrossRef]
- Zhao, B.; Lin, M.; Xiao, S.; Ouyang, J.; Al-Dhahir, N. IRS Empowered Robust Secure Transmission for Integrated Satellite-Terrestrial Networks. *IEEE Wirel. Commun. Lett.* 2023, 12, 336–340. [CrossRef]
- 8. Okogbaa, F.; Ahmed, Q.; Khan, F.; Abbas, W.; Che, F.; Zaidi, S.; Alade, T. Design and Application of Intelligent Reflecting Surface (IRS) for Beyond 5G Wireless Networks: A Review. *Sensor* 2023, 22, 2436. [CrossRef]
- 9. Luo, Y.; Ratnarajah, T.; Xue, J.; Khan, F.A. Interference Alignment in Two-Tier Randomly Distributed Heterogeneous Wireless Networks Using Stochastic Geometry Approach. *IEEE Syst. J.* 2018, *12*, 2238–2249. [CrossRef]
- 10. Qiu, J.; Grace, D.; Ding, G.; Zakaria, M.D.; Wu, Q. Air-Ground Heterogeneous Networks for 5G and Beyond via Integrating High and Low Altitude Platforms. *IEEE Wirel. Commun.* **2019**, *26*, 140–148. [CrossRef]
- Alam, S.; Kurt, G.K.; Yanikömeroglu, H.; Zhu, P.; Đào, N.D. High Altitude Platform Station Based Super Macro Base Station Constellations. *IEEE Commun. Mag.* 2021, 59, 103–109. [CrossRef]
- Kurt, G.K.; Khoshkholgh, M.G.; Alfattani, S.; Ibrahim, A.; Darwish, T.; Alam, S.; Yanikömeroglu, H.; Yongaçoglu, A. A Vision and Framework for the High Altitude Platform Station (HAPS) Networks of the Future. *IEEE Commun. Surv. Tutor.* 2021, 23, 729–779. [CrossRef]
- 13. Ye, J.; Qiao, J.; Kammoun, A.; Alouini, M.S. Nonterrestrial Communications Assisted by Reconfigurable Intelligent Surfaces. *Proc. IEEE* 2022, 110, 1423–1465. [CrossRef]
- 14. Azari, M.M.; Solanki, S.; Chatzinotas, S.; Kodheli, O.; Sallouha, H.; Colpaert, A.; Montoya, J.F.M.; Pollin, S.; Haqiqatnejad, A.; Mostaani, A.; et al. Evolution of Non-Terrestrial Networks From 5G to 6G: A Survey. *IEEE Commun. Surv. Tutor.* **2022**, 24, 2633–2672. [CrossRef]
- Wang, J.; Tang, W.; Han, Y.; Jin, S.; Xiao, L.; Wen, C.K.; Cheng, Q.; Cui, T.J. Interplay Between RIS and AI in Wireless Communications: Fundamentals, Architectures, Applications, and Open Research Problems. *IEEE J. Sel. Areas Commun.* 2021, 39, 2271–2288. [CrossRef]
- 16. Zhou, H.; Erol-Kantarci, M.; Liu, Y.; Poor, H.V. A Survey on Model-Based, Heuristic, and Machine Learning Optimization Approaches in RIS-Aided Wireless Networks. *IEEE Commun. Surv. Tutor.* **2023**, *26*, 781–823. [CrossRef]
- 17. Lou, Z.; Youcef Belmekki, B.E.; Alouini, M.S. HAPS in the Non-Terrestrial Network Nexus: Prospective Architectures and Performance Insights. *IEEE Wirel. Commun.* 2023, 30, 52–58. [CrossRef]

- Zhou, D.; Gao, S.; Liu, R.; Gao, F.; Guizani, M. Overview of development and regulatory aspects of high altitude platform system. *Intell. Converg. Netw.* 2020, 1, 58–78. [CrossRef]
- ElMossallamy, M.A.; Zhang, H.; Song, L.; Seddik, K.G.; Han, Z.; Li, G.Y. Reconfigurable Intelligent Surfaces for Wireless Communications: Principles, Challenges, and Opportunities. *IEEE Trans. Cogn. Commun. Netw.* 2020, 6, 990–1002. [CrossRef]
- Andrews, J.G.; Buzzi, S.; Choi, W.; Hanly, S.V.; Lozano, A.; Soong, A.C.; Zhang, J.C. What will 5G be? *IEEE J. Sel. Areas Commun.* 2014, 32, 1065–1082. [CrossRef]
- Ngo, H.Q.; Ashikhmin, A.; Yang, H.; Larsson, E.G.; Marzetta, T.L. Cell-free massive MIMO versus small cells. *IEEE Trans. Wirel. Commun.* 2017, 16, 1834–1850. [CrossRef]
- Ghosh, S.; Chakraborty, S.; Saha, D.; Ekpo, S.C.; Chakraborty, A.; Elias, F.; Uko, M. Design and Analysis of mm-Wave MIMO SIW Antenna for Multibeam 5G Applications. In Proceedings of the 2023 IEEE-APS Topical Conference on Antennas and Propagation in Wireless Communications (APWC), Venice, Italy, 9–13 October 2023; pp. 154–159. [CrossRef]
- 23. Nayebi, E.; Ashikhmin, A.; Marzetta, T.L.; Yang, H.; Rao, B.D. Precoding and Power Optimization in Cell-Free Massive MIMO Systems. *IEEE Trans. Wirel. Commun.* 2017, *16*, 4445–4459. [CrossRef]
- 24. Jeena, G.; Uko, M.; Ekpo, S.; Elias, F. Design of an elliptically-slotted patch antenna for multi-purpose wireless wi-Fi and biosensing applications. *e-Prime Adv. Electr. Eng. Electron. Energy* **2023**, *6*, 100368. [CrossRef]
- Lu, H.; Zeng, Y.; Jin, S.; Zhang, R. Enabling panoramic full-angle reflection via aerial intelligent reflecting surface. In Proceedings of the 2020 IEEE International Conference on Communications Workshops (ICC Workshops), Dublin, Ireland, 1–7 June 2020; pp. 1–6.
- 26. Dai, J.Y.; Zhao, J.; Cheng, Q.; Cui, T.J. Independent control of harmonic amplitudes and phases via a time-domain digital coding metasurface. *Light. Sci. Appl.* **2018**, *7*, 90. [CrossRef] [PubMed]
- Yu, X.; Xu, D.; Schober, R. MISO wireless communication systems via intelligent reflecting surfaces. In Proceedings of the 2019 IEEE/CIC International Conference on Communications in China (ICCC), Changchun, China, 11–13 August 2019; pp. 735–740.
- 28. Zhang, J.; Tao, D. Empowering things with intelligence: A survey of the progress, challenges, and opportunities in artificial intelligence of things. *IEEE Internet Things J.* **2020**, *8*, 7789–7817. [CrossRef]
- 29. Pan, C.; Ren, H.; Wang, K.; Xu, W.; Elkashlan, M.; Nallanathan, A.; Hanzo, L. Multicell MIMO communications relying on intelligent reflecting surfaces. *IEEE Trans. Wirel. Commun.* **2020**, *19*, 5218–5233. [CrossRef]
- Zhang, Y.; Di, B.; Zhang, H.; Lin, J.; Li, Y.; Song, L. Reconfigurable intelligent surface aided cell-free MIMO communications. *IEEE Wirel. Commun. Lett.* 2020, 10, 775–779. [CrossRef]
- Zhang, Z.; Dai, L. Capacity improvement in wideband reconfigurable intelligent surface-aided cell-free network. In Proceedings of the 2020 IEEE 21st International Workshop on Signal Processing Advances in Wireless Communications (SPAWC), Atlanta, GA, USA, 26–29 May 2020; pp. 1–5.
- 32. Alluhaibi, O.; Ahmed, Q.Z.; Kampert, E.; Higgins, M.D.; Wang, J. Revisiting the Energy-Efficient Hybrid D-A Precoding and Combining Design for mm-Wave Systems. *IEEE Trans. Green Commun. Netw.* **2020**, *4*, 340–354. [CrossRef]
- Nair, M.; Ahmed, Q.Z.; Zhu, H. Hybrid Digital-to-Analog Beamforming for Millimeter-Wave Systems with High User Density. In Proceedings of the 2016 IEEE Global Communications Conference (GLOBECOM), Washington, DC, USA, 4–8 December 2016; pp. 1–6. [CrossRef]
- Husbands, R.; Ahmed, Q.; Wang, J. Transmit antenna selection for massive MIMO: A knapsack problem formulation. In Proceedings of the 2017 IEEE International Conference on Communications (ICC), Paris, France, 21–25 May 2017; pp. 1–6. [CrossRef]
- 35. Ahmad, I.; Suomalainen, J.; Porambage, P.; Gurtov, A.; Huusko, J.; Höyhtyä, M. Security of Satellite-Terrestrial Communications: Challenges and Potential Solutions. *IEEE Access* **2022**, *10*, 96038–96052. [CrossRef]
- Abbasi, O.; Yadav, A.; Yanikömeroglu, H.; Đào, N.D.; Senarath, G.; Zhu, P. HAPS for 6G Networks: Potential Use Cases, Open Challenges, and Possible Solutions. *IEEE Wireless Commun.* 2024, 31, 324–331. [CrossRef]
- Sudheesh, P.G.; Magarini, M.; Muthuchidambaranathan, P. Achieving maximum system capacity in multiple-high altitude platforms through interference alignment. In Proceedings of the 2016 11th International Conference on Industrial and Information Systems (ICIIS), Roorkee, India, 3–4 December 2016; pp. 139–143. [CrossRef]
- Alfattani, S.; Jaafar, W.; Hmamouche, Y.; Yanikömeroglu, H.; Yongaçoglu, A.; Dào, N.D.; Zhu, P. Aerial Platforms with Reconfigurable Smart Surfaces for 5G and Beyond. *IEEE Commun. Mag.* 2021, 59, 96–102. [CrossRef]
- Zhao, J.; Chen, M.; Chen, M.; Yang, Z.; Wang, Y.; Cao, B.; Shikh-Bahaei, M. Energy Efficient Full-Duplex Communication Systems with Reconfigurable Intelligent Surface. In Proceedings of the 2020 IEEE 92nd Vehicular Technology Conference (VTC2020-Fall), Victoria, BC, Canada, 18 November–16 December 2020; pp. 1–5. [CrossRef]
- Ku, C.J.; Shen, L.H.; Feng, K.T. Reconfigurable Intelligent Surface-Empowered Self-Interference Cancellation for 6G Full-Duplex MIMO Communication Systems. arXiv 2022, arXiv:2112.07833.
- Yu, B.; Qian, C.; Lin, P.; Shao, S.; Pan, W.; Shen, Y.; Hu, S.; Su, D.; Sun, C.; Xiong, Q.; et al. Full Duplex Communication with Practical Self-Interference Cancellation Implementation. In Proceedings of the ICC 2022-IEEE International Conference on Communications, Seoul, Republic of Korea, 16–20 May 2022; pp. 1100–1105. [CrossRef]
- Luo, H.; Holm, M.; Ratnarajah, T. Design and Analysis of Wideband Self-Interference Cancellation for Full-Duplex Wireless Networks. In Proceedings of the 2022 IEEE Wireless Communications and Networking Conference (WCNC), Austin, TX, USA, 10–13 April 2022; pp. 2697–2702. [CrossRef]

- 43. Wang, Y.; Guan, P.; Yu, H.; Zhao, Y. Transmit Power Optimization of Simultaneous Transmission and Reflection RIS Assisted Full-Duplex Communications. *IEEE Access* **2022**, *10*, 61192–61200. [CrossRef]
- 44. Smida, B.; Sabharwal, A.; Fodor, G.; Alexandropoulos, G.C.; Suraweera, H.A.; Chae, C.B. Full-duplex wireless for 6G: Progress brings new opportunities and challenges. *IEEE J. Sel. Areas Commun.* **2023**, *41*, 2729–2750. [CrossRef]
- Khalid, W.; Yu, H.; Do, D.T.; Kaleem, Z.; Noh, S. RIS-Aided Physical Layer Security With Full-Duplex Jamming in Underlay D2D Networks. *IEEE Access* 2021, 9, 99667–99679. [CrossRef]
- Alluhaibi, O.; Ahmed, Q.Z.; Pan, C.; Zhu, H. Hybrid Digital-to-Analog Beamforming Approaches to Maximise the Capacity of mm-Wave Systems. In Proceedings of the 2017 IEEE 85th Vehicular Technology Conference (VTC Spring), Sydney, Australia, 4–7 June 2017; pp. 1–5. [CrossRef]
- Faisal, A.; Al-Nahhal, I.; Dobre, O.A.; Ngatched, T.M.N. On Discrete Phase Shifts Optimization of RIS-Aided FD Systems: Are All RIS Elements Needed? In Proceedings of the ICC 2023-IEEE International Conference on Communications, Rome, Italy, 28 May–1 June 2023; pp. 1382–1387. [CrossRef]
- 48. Guo, Y.; Liu, Y.; Wu, Q.; Li, X.; Shi, Q. Joint Beamforming and Power Allocation for RIS Aided Full-Duplex Integrated Sensing and Uplink Communication System. *IEEE Trans. Wirel. Commun.* **2023**, *23*, 4627–4642 [CrossRef]
- Björnson, E.; Sanguinetti, L. Power Scaling Laws and Near-Field Behaviors of Massive MIMO and Intelligent Reflecting Surfaces. *IEEE Open J. Commun. Soc.* 2020, 1, 1306–1324. [CrossRef]
- 50. Yang, D.; Xu, J.; Xu, W.; Huang, Y.; Lu, Z. Secure Communication for Spatially Correlated RIS-Aided Multiuser Massive MIMO Systems: Analysis and Optimization. *IEEE Commun. Lett.* **2023**, *27*, 797–801. [CrossRef]
- 51. Ahmed, Q.Z.; Ahmed, S.; Alouini, M.S.; Aïssa, S. Minimizing the Symbol-Error-Rate for Amplify-and-Forward Relaying Systems Using Evolutionary Algorithms. *IEEE Trans. Commun.* **2015**, *63*, 390–400. [CrossRef]
- 52. Ahmed, Q.Z.; Park, K.H.; Alouini, M.S.; Aïssa, S. Compression and Combining Based on Channel Shortening and Reduced-Rank Techniques for Cooperative Wireless Sensor Networks. *IEEE Trans. Veh. Technol.* **2014**, *63*, 72–81. [CrossRef]
- Chougrani, H.; Kisseleff, S.; Martins, W.A.; Chatzinotas, S. NB-IoT Random Access for Nonterrestrial Networks: Preamble Detection and Uplink Synchronization. *IEEE Internet Things J.* 2022, *9*, 14913–14927. [CrossRef]
- 54. Dazhi, M.N.; Al-Hraishawi, H.; Shankar, M.R.B.; Chatzinotas, S.; Ottersten, B. Energy-Efficient Service-Aware Multi-Connectivity Scheduler for Uplink Multi-Layer Non-Terrestrial Networks. *IEEE Trans. Green Commun. Netw.* 2023, 7, 1326–1341. [CrossRef]
- 55. Jamshed, M.A.; Kaushik, A.; Toka, M.; Shin, W.; Shakir, M.Z.; Dash, S.P.; Dardari, D. Synergizing Airborne Non-Terrestrial Networks and Reconfigurable Intelligent Surfaces-Aided 6G IoT. *IEEE Internet Things Mag.* **2024**, *7*, 46–52. [CrossRef]
- 56. Sharma, V.; Yaswanth, J.; Singh, S.K.; Biswas, S.; Singh, K.; Khan, F. A Pricing-Based Approach for Energy-Efficiency Maximization in RIS-Aided Multi-User MIMO SWIPT-Enabled Wireless Networks. *IEEE Access* **2022**, *10*, 29132–29148. [CrossRef]
- Karim, F.; Singh, S.K.; Singh, K.; Khan, F. STAR-RIS-aided Full Duplex Communications with FBL Transmission. In Proceedings of the 2023 IEEE Wireless Communications and Networking Conference (WCNC), Glasgow, UK, 26–29 March 2023; pp. 1–6. [CrossRef]
- Basar, E.; Renzo, M.D.; de Rosny, J.; Debbah, M.; Alouini, M.S.; Zhang, R. Wireless Communications Through Reconfigurable Intelligent Surfaces. *IEEE Access* 2019, 7, 116753–116773. [CrossRef]
- 59. Liu, Y.; Liu, X.; Mu, X.; Hou, T.; Xu, J.; Renzo, M.D.; Al-Dhahir, N. Reconfigurable Intelligent Surfaces: Principles and Opportunities. *IEEE Commun. Surv. Tutor.* 2021, 23, 1546–1577. [CrossRef]
- 60. Rasilainen, K.; Phan, T.T.; Berg, M.; Pärssinen, A.; Soh, P.J. Hardware Aspects of Sub-THz Antennas and Reconfigurable Intelligent Surfaces for 6G Communications. *IEEE J. Sel. Areas Commun.* **2023**, *41*, 2530–2546. [CrossRef]
- Ahmed, Q.Z.; Yang, L.L. Performance of Hybrid Direct-Sequence Time-Hopping Ultrawide Bandwidth Systems in Nakagami-M Fading Channels. In Proceedings of the 2007 IEEE 18th International Symposium on Personal, Indoor and Mobile Radio Communications, Athens, Greece, 3–7 September 2007; pp. 1–5. [CrossRef]
- 62. Wu, Q.; Zhang, R. Intelligent Reflecting Surface Enhanced Wireless Network via Joint Active and Passive Beamforming. *IEEE Trans. Wirel. Commun.* **2019**, *18*, 5394–5409. [CrossRef]
- 63. Wu, Q.; Zhang, R. Towards Smart and Reconfigurable Environment: Intelligent Reflecting Surface Aided Wireless Network. *IEEE Commun. Mag.* 2020, *58*, 106–112. [CrossRef]
- 64. Tang, X.; He, H.; Dong, L.; Li, L.; Du, Q.; Han, Z. Robust Secrecy via Aerial Reflection and Jamming: Joint Optimization of Deployment and Transmission. *IEEE Internet Things J.* 2023, 10, 12562–12576. [CrossRef]
- 65. Huang, C.; Zappone, A.; Alexandropoulos, G.C.; Debbah, M.; Yuen, C. Reconfigurable Intelligent Surfaces for Energy Efficiency in Wireless Communication. *IEEE Trans. Wirel. Commun.* **2019**, *18*, 4157–4170. [CrossRef]
- Wu, Q.; Zhang, S.; Zheng, B.; You, C.; Zhang, R. Intelligent Reflecting Surface-Aided Wireless Communications: A Tutorial. *IEEE Trans. Commun.* 2021, 69, 3313–3351. [CrossRef]
- Azizi, A.; Farhang, A. RIS Meets Aerodynamic HAPS: A Multi-Objective Optimization Approach. *IEEE Wirel. Commun. Lett.* 2023, 12, 1851–1855. [CrossRef]
- Renzo, M.D.; Zappone, A.; Debbah, M.; Alouini, M.S.; Yuen, C.; de Rosny, J.; Tretyakov, S. Smart Radio Environments Empowered by Reconfigurable Intelligent Surfaces: How It Works, State of Research, and The Road Ahead. *IEEE J. Sel. Areas Commun.* 2020, 38, 2450–2525. [CrossRef]
- 69. Uko, M.; Ekpo, S. 8-12 GHz pHEMT MMIC low-noise amplifier for 5G and fiber-integrated satellite applications. *Int. Rev. Aerosp. Eng. (IREASE)* **2020**, *13*, 99–107. [CrossRef]

- Huang, H. FDMA with Layers-based Optimized Mobile Relays Subsets Algorithm in B5G/6G Cognitive IoT Networks. *arXiv* 2020, arXiv:2005.01384.
- Charania, I.; Li, X. Smart farming: Agriculture's shift from a labor intensive to technology native industry. *Internet Things* 2020, 9, 100142. [CrossRef]
- 72. Yao, Y.; Lv, K.; Ma, N.; Yue, X.; Qin, X.; Yun, X. Energy efficient air-to-ground communication networks with reconfigurable intelligent surface. *J. Commun. Netw.* **2022**, *24*, 555–565. [CrossRef]
- Guo, C.; Cui, Y.; Yang, F.; Ding, L. Outage probability analysis and minimization in intelligent reflecting surface-assisted MISO systems. *IEEE Commun. Lett.* 2020, 24, 1563–1567. [CrossRef]
- You, C.; Zhang, R. 3D trajectory optimization in Rician fading for UAV-enabled data harvesting. *IEEE Trans. Wirel. Commun.* 2019, 18, 3192–3207. [CrossRef]
- He, Z.Q.; Yuan, X. Cascaded channel estimation for large intelligent metasurface assisted massive MIMO. *IEEE Wirel. Commun. Lett.* 2019, 9, 210–214. [CrossRef]
- Njima, W.; Chafii, M.; Chorti, A.; Shubair, R.M.; Poor, H.V. Indoor localization using data augmentation via selective generative adversarial networks. *IEEE Access* 2021, 9, 98337–98347. [CrossRef]
- 77. Sheen, B.; Yang, J.; Feng, X.; Chowdhury, M.M.U. A digital twin for reconfigurable intelligent surface assisted wireless communication. *arXiv* **2020**, arXiv:2009.00454.
- 78. González, E.Y.; Medina, J.A.; Murillo, J.G. High sensitivity photonic crystal sensor based on transition between photonic bands. *Laser Phys.* **2022**, *32*, 106202. [CrossRef]
- Geraci, G.; Garcia-Rodriguez, A.; Azari, M.M.; Lozano, A.; Mezzavilla, M.; Chatzinotas, S.; Chen, Y.; Rangan, S.; Di Renzo, M. What will the future of UAV cellular communications be? A flight from 5G to 6G. *IEEE Commun. Surv. Tutor.* 2022, 24, 1304–1335. [CrossRef]
- Zheng, Z.; Jing, W.; Lu, Z.; Wen, X. RIS-Enhanced LEO Satellite Communication: Joint Passive Beamforming and Orientation Optimization. In Proceedings of the 2022 IEEE Globecom Workshops (GC Wkshps), Rio de Janeiro, Brazil, 4–8 December 2022; pp. 874–879. [CrossRef]
- 81. Tang, J.; Bian, D.; Li, G.; Hu, J.; Cheng, J. Resource Allocation for LEO Beam-Hopping Satellites in a Spectrum Sharing Scenario. *IEEE Access* 2021, 9, 56468–56478. [CrossRef]
- Liu, R.; Li, M.; Liu, Y.; Wu, Q.; Liu, Q. Joint Transmit Waveform and Passive Beamforming Design for RIS-Aided DFRC Systems. IEEE J. Sel. Top. Signal Process. 2022, 16, 995–1010. [CrossRef]
- 83. Zhao, Y.; Clerckx, B.; Feng, Z. IRS-aided SWIPT: Joint waveform, active and passive beamforming design under nonlinear harvester model. *IEEE Trans. Commun.* 2021, 70, 1345–1359. [CrossRef]
- Elsaraf, Z.; Khan, F.A.; Ahmed, Q.Z. Deep Learning Based Power Allocation Schemes in NOMA Systems: A Review. In Proceedings of the 2021 26th International Conference on Automation and Computing (ICAC), Portsmouth, UK, 2–4 September 2021; pp. 1–6. [CrossRef]
- 85. Riaz, S.; Khan, F.A.; Saleem, S.; Ahmed, Q.Z. Reducing the Mutual Outage Probability of Cooperative Non-Orthogonal Multiple Access. *IEEE Trans. Veh. Technol.* 2020, 69, 16207–16212. [CrossRef]
- Ahmed, A.; Elsaraf, Z.; Khan, F.A.; Ahmed, Q.Z. Cooperative Non-Orthogonal Multiple Access for Beyond 5G Networks. *IEEE Open J. Commun. Soc.* 2021, 2, 990–999. [CrossRef]
- Elsaraf, Z.; Ahmed, A.; Khan, F.A.; Ahmed, Q.Z. EXIT Chart Analysis of Cooperative Non-Orthogonal Multiple Access for Next Generation Wireless Communication Systems. In Proceedings of the 2020 European Conference on Networks and Communications (EuCNC), Dubrovnik, Croatia, 15–18 June 2020; pp. 281–285.
- 88. Elsaraf, Z.; Ahmed, A.; Khan, F.A.; Ahmed, Q.Z. Cooperative non-orthogonal multiple access for wireless communication networks by exploiting the EXIT chart analysis. *EURASIP J. Wirel. Commun. Netw.* **2021**, *7*, 990–999. [CrossRef]
- Khennoufa, F.; Abdellatif, K.; Kara, F.; Yanikomeroglu, H.; Rabie, K.; Elganimi, T.Y.; Beddiaf, S. Error Performance Analysis of UAV-Mounted RIS for NOMA Systems With Practical Constraints. *IEEE Commun. Lett.* 2024, 28, 887–891. [CrossRef]
- Lin, Z.; Lin, M.; Champagne, B.; Zhu, W.P.; Al-Dhahir, N. Secrecy-Energy Efficient Hybrid Beamforming for Satellite-Terrestrial Integrated Networks. *IEEE Trans. Commun.* 2021, 69, 6345–6360. [CrossRef]
- Xie, Y.; Hu, Y.; Chen, Y.; Liu, Y.; Shou, G. A video analytics-based intelligent indoor positioning system using edge computing for IoT. In Proceedings of the 2018 International Conference on Cyber-Enabled Distributed Computing and Knowledge Discovery (CyberC), Zhengzhou, China, 18–20 October 2018; pp. 118–1187.
- 92. Khan, M.S.; Iftikhar, A.; Shubair, R.M.; Capobianco, A.D.; Braaten, B.D.; Anagnostou, D.E. Ultra-wideband antenna with MIMO diversity for 5G wireless communication. *arXiv* 2020, arXiv:2007.02294.
- Alam, A.; Ullah, I.; Lee, Y.K. Video big data analytics in the cloud: A reference architecture, survey, opportunities, and open research issues. *IEEE Access* 2020, *8*, 152377–152422. [CrossRef]
- Jang, S.Y.; Lee, Y.; Shin, B.; Lee, D. Application-aware IoT camera virtualization for video analytics edge computing. In Proceedings of the 2018 IEEE/ACM Symposium on Edge Computing (SEC), Seattle, WA, USA, 25–27 October 2018; pp. 132–144.
- 95. Tekbiyik, K.; Kurt, G.K.; Ekti, A.R.; Yanikomeroglu, H. Reconfigurable intelligent surfaces in action for nonterrestrial networks. *IEEE Veh. Technol. Mag.* **2022**, *17*, 45–53. [CrossRef]
- 96. Ghosal, S.; De, A.; Shubair, R.M.; Chakrabarty, A. Near-Field Radiation Exposure Control in Slot-Loaded Microstrip Antenna: A Characteristic Mode Approach. *arXiv* 2019, arXiv:1907.12127.

- 97. Hashida, H.; Kawamoto, Y.; Kato, N. Intelligent Reflecting Surface Placement Optimization in Air-Ground Communication Networks Toward 6G. *IEEE Wirel. Commun.* 2020, 27, 146–151. [CrossRef]
- Feng, H.; Xu, Y.; Zhao, Y. Deep Learning-Based Joint Channel Estimation and CSI Feedback for RIS-Assisted Communications. IEEE Commun. Lett. 2024, 28, 1860–1864. [CrossRef]
- 99. Jeong, S.; Farhang, A.; Perović, N.S.; Flanagan, M.F. Joint CFO and Channel Estimation for RIS-Aided Multi-User Massive MIMO Systems. *IEEE Trans. Veh. Technol.* 2023, 72, 11800–11813. [CrossRef]
- Seo, J.; Choi, G.; Kim, S.C. DBPN-Based Uplink Channel Estimation for Multi-User MISO RIS System. *IEEE Wirel. Commun. Lett.* 2023, 12, 2143–2147. [CrossRef]
- 101. Dahrouj, H.; Liu, S.; Alouini, M.S. Machine Learning-Based User Scheduling in Integrated Satellite-HAPS-Ground Networks. *IEEE Netw.* 2023, 37, 102–109. [CrossRef]
- 102. Chen, W.; Lin, X.; Lee, J.; Toskala, A.; Sun, S.; Chiasserini, C.F.; Liu, L. 5G-Advanced Toward 6G: Past, Present, and Future. *IEEE J. Sel. Areas Commun.* 2023, *41*, 1592–1619. [CrossRef]
- 103. Sharma, T.; Chehri, A.; Fortier, P. Reconfigurable intelligent surfaces for 5G and beyond wireless communications: A comprehensive survey. *Energies* **2021**, *14*, 8219. [CrossRef]
- 104. Wang, C.X.; You, X.; Gao, X.; Zhu, X.; Li, Z.; Zhang, C.; Wang, H.; Huang, Y.; Chen, Y.; Haas, H.; et al. On the road to 6G: Visions, requirements, key technologies and testbeds. *IEEE Commun. Surv. Tutor.* 2023, 25, 905–974. [CrossRef]
- 105. Letaief, K.B.; Shi, Y.; Lu, J.; Lu, J. Edge artificial intelligence for 6G: Vision, enabling technologies, and applications. *IEEE J. Sel. Areas Commun.* **2021**, 40, 5–36. [CrossRef]
- 106. Enahoro, S.; Ekpo, S.; Uko, M.; Alabi, S.; Elias, F.; Unnikrishnan, R. A Metamaterial-Grounded Ultra-Wideband Cross-Fractal MIMO Antenna for K, Ka, and mmWave Applications. *e-Prime Adv. Electr. Eng. Electron. Energy* 2023, 1–19. [CrossRef]
- 107. Garg, N.; Sellathurai, M.; Khan, F.; Ratnarajah, T.; Bhatia, V. Success Probability Analysis for Edge Caching in Massive MIMO Networks. In Proceedings of the 2019 IEEE International Conference on Advanced Networks and Telecommunications Systems (ANTS), Goa, India, 16–19 December 2019; pp. 1–5.
- Liu, Q.; Feng, Z.; Chen, D.; Tan, F.; He, C. Empowering 6G Non-Terrestrial Networks With Intelligent Reflection Technologies for IoT Applications. *IEEE Netw.* 2024, 38, 96–102. [CrossRef]
- 109. Kaushik, A.; Singh, R.; Li, M.; Luo, H.; Dayarathna, S.; Senanayake, R.; An, X.; Stirling-Gallacher, R.A.; Shin, W.; Di Renzo, M. Integrated Sensing and Communications for IoT: Synergies with Key 6G Technology Enablers. *IEEE Internet Things Mag.* 2024, 7, 136–143. [CrossRef]
- Liu, R.; Wu, Q.; Di Renzo, M.; Yuan, Y. A path to smart radio environments: An industrial viewpoint on reconfigurable intelligent surfaces. *IEEE Wirel. Commun.* 2022, 29, 202–208. [CrossRef]
- Shojaeifard, A.; Wong, K.K.; Tong, K.F.; Chu, Z.; Mourad, A.; Haghighat, A.; Hemadeh, I.; Nguyen, N.T.; Tapio, V.; Juntti, M. MIMO evolution beyond 5G through reconfigurable intelligent surfaces and fluid antenna systems. *Proc. IEEE* 2022, 110, 1244–1265. [CrossRef]
- 112. Sarawar, A.B.; Badrudduza, A.S.M.; Ibrahim, M.; Ansari, I.S.; Yu, H. Secrecy Performance Analysis of Integrated RF-UOWC IoT Networks Enabled by UAV and Underwater-RIS. *IEEE Internet Things J.* **2024**. [CrossRef]
- 113. Pan, C.; Zhou, G.; Zhi, K.; Hong, S.; Wu, T.; Pan, Y.; Ren, H.; Di Renzo, M.; Swindlehurst, A.L.; Zhang, R.; et al. An overview of signal processing techniques for RIS/IRS-aided wireless systems. *IEEE J. Sel. Top. Signal Process.* 2022, *16*, 883–917. [CrossRef]
- Yousif, E.; Khan, F.; Ratnarajah, T.; Sellathurai, M. On the spectral coexistence of colocated MIMO radars and wireless communications systems. In Proceedings of the 2016 IEEE 17th International Workshop on Signal Processing Advances in Wireless Communications (SPAWC), Edinburgh, UK, 3–6 July 2016; pp. 1–5.
- Yousif, E.H.G.; Filippou, M.C.; Khan, F.; Ratnarajah, T.; Sellathurai, M. A new LSA-based approach for spectral coexistence of MIMO radar and wireless communications systems. In Proceedings of the 2016 IEEE International Conference on Communications (ICC), Kuala Lumpur, Malaysia, 23–27 May 2016; pp. 1–6.
- Uko, M.; Ekpo, S. A 23-28 GHz pHEMT MMIC Low-Noise Amplifier for Satellite-Cellular Convergence Applications. Int. Rev. Aerosp. Eng. J. 2021, 14, 1–10. [CrossRef]
- 117. Zhang, M.; Cumanan, K.; Thiyagalingam, J.; Wang, W.; Burr, A.G.; Ding, Z.; Dobre, O.A. Energy Efficiency Optimization for Secure Transmission in MISO Cognitive Radio Network With Energy Harvesting. *IEEE Access* 2019, 7, 126234–126252. [CrossRef]
- 118. Chen, Z.; Shi, Q.; Wu, Q.; Xu, W. Joint Transceiver Optimization of MIMO SWIPT Systems for Harvested Power Maximization. *IEEE Signal Process. Lett.* **2017**, 24, 1557–1561. [CrossRef]
- Yu, X.; Teng, T.; Dang, X.; Leung, S.H.; Xu, F. Joint Power Allocation and Beamforming for Energy-Efficient Design in Multiuser Distributed MIMO Systems. *IEEE Trans. Commun.* 2021, 69, 4128–4143. [CrossRef]
- Huo, Y.; Dong, X.; Ferdinand, N. Distributed Reconfigurable Intelligent Surfaces for Energy-Efficient Indoor Terahertz Wireless Communications. *IEEE Internet Things J.* 2023, 10, 2728–2742. [CrossRef]
- 121. He, J.; Mao, Y.; Zhou, Y.; Wang, T.; Shi, Y. Reconfigurable Intelligent Surfaces Empowered Green Wireless Networks With User Admission Control. *IEEE Trans. Commun.* 2023, 71, 4062–4078. [CrossRef]
- 122. Xie, H.; Li, D. To Reflect or Not to Reflect: On–Off Control and Number Configuration for Reflecting Elements in RIS-Aided Wireless Systems. *IEEE Trans. Commun.* 2023, 71, 7409–7424. [CrossRef]

- 123. Yang, Y.; Gursoy, M.C. Energy-Efficient Scheduling in RIS-aided MEC Networks with NOMA and Finite Blocklength Codes. In Proceedings of the 2022 International Symposium on Wireless Communication Systems (ISWCS), Hangzhou, China, 19–22 October 2022; pp. 1–6. [CrossRef]
- 124. Yu, Z.; Yuan, D. Resource Optimization With Interference Coupling in Multi-RIS-Assisted Multi-Cell Systems. *IEEE Open J. Veh. Technol.* 2022, *3*, 98–110. [CrossRef]
- 125. Huang, J.; Wang, C.X.; Sun, Y.; Feng, R.; Huang, J.; Guo, B.; Zhong, Z.; Cui, T.J. Reconfigurable intelligent surfaces: Channel characterization and modeling. *Proc. IEEE* 2022, *110*, 1290–1311. [CrossRef]
- 126. Das, S.K.; Benkhelifa, F.; Sun, Y.; Abumarshoud, H.; Abbasi, Q.H.; Imran, M.A.; Mohjazi, L. Comprehensive review on ML-based RIS-enhanced IoT systems: Basics, research progress and future challenges. *Comput. Netw.* **2023**, 224, 109581. [CrossRef]
- 127. Ren, H.; Chen, Z.; Hu, G.; Peng, Z.; Pan, C.; Wang, J. Transmission Design for Active RIS-Aided Simultaneous Wireless Information and Power Transfer. *IEEE Wirel. Commun. Lett.* **2023**, *12*, 600–604. [CrossRef]
- 128. Xie, H.; Gu, B.; Li, D.; Lin, Z.; Xu, Y. Gain Without Pain: Recycling Reflected Energy From Wireless-Powered RIS-Aided Communications. *IEEE Internet Things J.* 2023, 10, 13264–13280. [CrossRef]
- 129. Zappone, A.; Renzo, M.D. Energy Efficiency Optimization of Reconfigurable Intelligent Surfaces With Electromagnetic Field Exposure Constraints. *IEEE Signal Process. Lett.* **2022**, *29*, 1447–1451. [CrossRef]
- Jeon, H.B.; Park, S.H.; Park, J.; Huang, K.; Chae, C.B. An Energy-Efficient Aerial Backhaul System With Reconfigurable Intelligent Surface. *IEEE Trans. Wirel. Commun.* 2022, 21, 6478–6494. [CrossRef]
- 131. Gao, Y.; Xiao, Y.; Lei, X.; Zhu, Q.; Niyato, D.; Wong, K.K.; Fan, P.; Hu, R.Q. Pricing for Reconfigurable Intelligent Surface Aided Wireless Networks: Models and Principles. *IEEE Netw.* **2023**, *37*, 102–110. [CrossRef]
- Liu, R.; Alexandropoulos, G.C.; Wu, Q.; Jian, M.; Liu, Y. How Can Reconfigurable Intelligent Surfaces Drive 5G-Advanced Wireless Networks: A Standardization Perspective. In Proceedings of the 2022 IEEE/CIC International Conference on Communications in China (ICCC Workshops), Foshan, China, 11–13 August 2022; pp. 221–226. [CrossRef]
- 133. Pan, C.; Ren, H.; Wang, K.; Kolb, J.F.; Elkashlan, M.; Chen, M.; Di Renzo, M.; Hao, Y.; Wang, J.; Swindlehurst, A.L.; et al. Reconfigurable Intelligent Surfaces for 6G Systems: Principles, Applications, and Research Directions. *IEEE Commun. Mag.* 2021, 59, 14–20. [CrossRef]
- Keşir, S.; Yağan, M.Y.; Hökelek, I.; Pusane, A.E.; Görçin, A. Rapid CNN-Assisted Iterative RIS Element Configuration. In Proceedings of the 2023 International Symposium on Networks, Computers and Communications (ISNCC), Doha, Qatar, 23–26 October 2023; pp. 1–6. [CrossRef]
- 135. Faisal, K.M.; Choi, W. Machine Learning Approaches for Reconfigurable Intelligent Surfaces: A Survey. *IEEE Access* 2022, 10, 27343–27367. [CrossRef]
- 136. Huang, S.; Wang, S.; Wang, R.; Wen, M.; Huang, K. Reconfigurable Intelligent Surface Assisted Mobile Edge Computing With Heterogeneous Learning Tasks. *IEEE Trans. Cogn. Commun. Netw.* **2021**, *7*, 369–382. [CrossRef]
- Pradhan, A.; Dhillon, H.S. A Probabilistic Reformulation Technique for Discrete RIS Optimization in Wireless Systems. *IEEE Trans. Wirel. Commun.* 2023, 23, 5377–5392. [CrossRef]
- Ahmed, Q.Z.; Hafeez, M.; Khan, F.A.; Lazaridis, P. Towards Beyond 5G Future Wireless Networks with focus towards Indoor Localization. In Proceedings of the 2020 IEEE Eighth International Conference on Communications and Networking (ComNet), Piscataway, NJ, USA, 27–30 October 2020; pp. 1–5. [CrossRef]
- Peng, B.; Besser, K.L.; Raghunath, R.; Jamali, V.; Jorswieck, E.A. RISnet: A Scalable Approach for Reconfigurable Intelligent Surface Optimization with Partial CSI. In Proceedings of the GLOBECOM 2023-2023 IEEE Global Communications Conference, Kuala Lumpur, Malaysia, 4–8 December 2023; pp. 4810–4816. [CrossRef]
- Yang, H.; Xie, X.; Kadoch, M. Machine Learning Techniques and A Case Study for Intelligent Wireless Networks. *IEEE Netw.* 2020, 34, 208–215. [CrossRef]
- 141. Wang, Y.; Ye, Z.; Wan, P.; Zhao, J. A survey of dynamic spectrum allocation based on reinforcement learning algorithms in cognitive radio networks. *Artif. Intell. Rev.* 2019, *51*, 493–506. [CrossRef]

Disclaimer/Publisher's Note: The statements, opinions and data contained in all publications are solely those of the individual author(s) and contributor(s) and not of MDPI and/or the editor(s). MDPI and/or the editor(s) disclaim responsibility for any injury to people or property resulting from any ideas, methods, instructions or products referred to in the content.